Author’s response to reviews

Title: Disruption of microbial cell morphology by Buxus macowanii

Authors:
Brian Ngobeni (ngobenibrian1@gmail.com)
Samson Sitheni Mashele (smashele@cut.ac.za)
Ntsoaki Joyce Malebo (nmalebo@cut.ac.za)
Elmarie van Der Watt (vdWattE@ufs.ac.za)
Idah Tichaidza Manduna (imanduna@cut.ac.za)

Version: 2 Date: 11 Jul 2020

Author’s response to reviews:

10 July 2020

To the Editor
BMC Complementary Medicine and Therapies

RE: Response to Editor’s comments
"Disruption of microbial cell morphology by Buxus macowanii" (BCAM-D-19-01810R1)

Comment:
The plagiarism detection software used by BMC Complementary Medicine and Therapies
detected a high level of similarity (46%) with published papers and documents. In fact, I examined
the software report noticed stretches identical to sentences present in other articles. Please review
the work again.

Response:
Thank you for the comment which was unexpected to us. Plagiarism is a serious matter and we
strive to avoid it as much as possible. We have revised our manuscript and submitted it to the
software that our institution provides and scored 9% which was categorised as low risk on
SafeAssign.

We did not get a response from the Editor to our previous request for your software’s originality
report.

Thank you and regards,
Dr Idah T Manduna